Author's response to reviews

Title: Development of Buffalo Hump in course of antiretroviral therapy including Raltegravir and unboosted Atazanavir: a case report and review of the literature.

Authors:

Giancarlo Ceccarelli (giancarlo.ceccarelli@uniroma1.it)
Gabriella d'Ettorre (g.ishtar@libero.it)
Francesco Marchetti (marchettifr@gmail.com)
Cecilia Rizza (rceci@libero.it)
Claudio M Mastroianni (claudio.mastroanni@uniroma1.it)
Bruno Carlesimo (carlesimbr@gmail.com)
Vincenzo Vullo (vincenzo.vullo@uniroma1.it)

Version: 3 Date: 9 November 2010

Author's response to reviews:

Cover letter

Dear Sir, we modified our manuscript as follow:

Affiliations - Each affiliation was written in full, in the following format: department, institute, city, and country. We changed Department of Tropical and Infectious Diseases in Department of Infectious Diseases and Public Health

Patient information: we wrote the ethnicity of the patient (Caucasian) in the abstract and case presentation sections of the manuscript.

Author contributions: we indicated that 'all authors have read and approved the final manuscript'.

Manuscript format: we removed the text underlining from the manuscript.

Best regards

Giancarlo Ceccarelli, MD PhD